Author's response to reviews

Title: Evaluation of dendritic cells loaded with apoptotic cancer cells or expressing tumour mRNA as potential cancer vaccines against leukemia

Authors:

Silvija Jarnjak-Jankovic (silvija.jankovic@klinmed.uio.no)
Rolf D Pettersen (rolf.pettersen@klinmed.uio.no)
Stein Saeboe-Larssen (stein.saboe-larssen@samfunnsmed.uio.no)
Finn Wesenberg (finn.wesenberg@rikshospitalet.no)
Gustav Gaudernack (gustav.gaudernack@labmed.uio.no)

Version: 4 Date: 10 February 2005

Author's response to reviews: see over
Dear Editor,

in manuscript I made next changes:
1. Keywords section - removed from the manuscript.
2. Abstract - structured into Background, Methods, Results, Conclusions.
3. Introduction - renamed as 'Background'.
4. Materials & Methods - renamed as 'Methods'.
5. Conclusions - insert after the Discussion.
6. Competing interests - included below he Conclusions.
7. Figure legends- These are now placed after the references